

LONDON
SCHOOL of
HYGIENE
& TROPICAL
MEDICINE



LSHTM Research Online

Asante, AD; Price, J; Hayen, A; Irava, W; Martins, J; Guinness, L; Ataguba, JE; Limwattananon, S; Mills, A; Jan, S; +1 more... Wiseman, V; (2014) Assessment of equity in healthcare financing in Fiji and Timor-Leste: a study protocol. *BMJ open*, 4 (12). e006806. ISSN 2044-6055 DOI: <https://doi.org/10.1136/bmjopen-2014-006806>

Downloaded from: <http://researchonline.lshtm.ac.uk/2026658/>

DOI: <https://doi.org/10.1136/bmjopen-2014-006806>

Usage Guidelines:

Please refer to usage guidelines at <https://researchonline.lshtm.ac.uk/policies.html> or alternatively contact researchonline@lshtm.ac.uk.

Available under license: <http://creativecommons.org/licenses/by-nc-nd/2.5/>

<https://researchonline.lshtm.ac.uk>

PEER REVIEW HISTORY

BMJ Open publishes all reviews undertaken for accepted manuscripts. Reviewers are asked to complete a checklist review form (<http://bmjopen.bmj.com/site/about/resources/checklist.pdf>) and are provided with free text boxes to elaborate on their assessment. These free text comments are reproduced below.

ARTICLE DETAILS

TITLE (PROVISIONAL)	Assessment of equity in health care financing in Fiji and Timor-Leste: a study protocol
AUTHORS	Asante, Augustine; Price, Jennifer; Hayen, Andrew; Irava, Wayne; Martins, Joao; Guinness, Lorna; Ataguba, John; Limwattananon, Supon; Mills, Anne; Jan, Stephen; Wiseman, Virginia

This article was not externally reviewed at BMJ Open. Protocols that have been independently assessed prior to submission to BMJ Open are usually fast-tracked to publication on the grounds that further substantial changes will not be possible. This independent assessment will usually be external, independent review for both a substantial grant award from a non-commercial or government funder and ethics approval.